## Routine gastric residual volume measurement to guide enteral feeding in mechanically ventilated infants and children: the GASTRIC feasibility study

Lyvonne N Tume,<sup>1\*</sup> Kerry Woolfall,<sup>2</sup> Barbara Arch,<sup>3</sup> Louise Roper,<sup>2</sup> Elizabeth Deja,<sup>2</sup> Ashley P Jones,<sup>3</sup> Lynne Latten,<sup>4</sup> Nazima Pathan,<sup>5</sup> Helen Eccleson,<sup>3</sup> Helen Hickey,<sup>3</sup> Roger Parslow,<sup>6</sup> Jennifer Preston,<sup>7</sup> Anne Beissel,<sup>8</sup> Izabela Andrzejewska,<sup>9</sup> Chris Gale,<sup>10</sup> Frederic V Valla<sup>11</sup> and Jon Dorling<sup>12</sup>

<sup>1</sup>School of Health and Society, University of Salford, Salford, UK

<sup>2</sup>Department of Health Services Research, University of Liverpool, Liverpool, UK <sup>3</sup>Liverpool Clinical Trials Centre, University of Liverpool, Liverpool, UK

- <sup>4</sup>Nutrition and Dietetics, Alder Hey Children's NHS Foundation Trust, Liverpool, UK <sup>5</sup>Department of Paediatrics, University of Cambridge, Cambridge, UK
- <sup>6</sup>School of Medicine, University of Leeds, Leeds, UK
- <sup>7</sup>Department of Women's and Children's Health, Institute of Translational Medicine (Child Health), Alder Hey Children's NHS Foundation Trust, University of Liverpool, Liverpool, UK
- <sup>8</sup>Neonatal Unit, Hôpital Femme Mère Enfant, Hospices Civils de Lyon, Lyon-Bron, France
- <sup>9</sup>Neonatal Unit, Chelsea and Westminster Hospital, London, UK
- <sup>10</sup>Neonatal Medicine, School of Public Health, Faculty of Medicine, Imperial College London, Chelsea and Westminster Hospital, London, UK
- <sup>11</sup>Paediatric Intensive Care Unit, Hôpital Femme Mère Enfant, Hospices Civils de Lyon, Lyon-Bron, France
- <sup>12</sup>Department of Pediatrics, Division of Neonatal-Perinatal Medicine, Faculty of Medicine, Dalhousie University, Halifax, NS, Canada

\*Corresponding author l.n.tume@salford.ac.uk

**Declared competing interests of authors:** Lyvonne N Tume was a National Institute for Health Research (NIHR) Health Technology Assessment (HTA) panel member during the conduct of the study and is the deputy chairperson of the HTA Prioritisation Committee. Chris Gale reports grants from the Medical Research Council during the conduct of the study; and grants from the NIHR (research grant and fellowship for a Doctor of Philosophy student), the Mason Medical Research Foundation (London, UK), Rosetrees Trust (Edgware, UK) and from the Canadian Institute for Health Research (Ottawa, ON, Canada), outside the submitted work. Chris Gale also reports grants and personal fees from Chiesi Pharmaceuticals (Parma, Italy), outside the submitted work (the grant is for a research study and the personal fee was to support attendance at an educational meeting). Chris Gale is vice-chairperson of the NIHR Research for Patient Benefit London Regional Assessment Panel (2016–present). Frederic V Valla reports personal fees from Baxter International (Deerfield, IL, USA) and personal fees from Nutricia (Zoetermeer, the Netherlands), outside the submitted work. Jon Dorling reports grants from the NIHR and from Nutrinia (Nazareth, Israel), outside the submitted work (the grant from Nutrinia in 2018 was for part of his salary to work as an expert advisor on a trial). Jon Dorling was a member of the NIHR HTA General Board (2017–18) and the NIHR HTA Maternity, Neonatal and Child Health Panel (2013–18).

**Disclaimer:** This report contains transcripts of interviews conducted in the course of the research and contains language that may offend some readers.

Published May 2020 DOI: 10.3310/hta24230

## **Plain English summary**

The GASTRIC feasibility study Health Technology Assessment 2020; Vol. 24: No. 23 DOI: 10.3310/hta24230

NIHR Journals Library www.journalslibrary.nihr.ac.uk

# **Plain English summary**

N urses looking after babies and children on intensive care units in the UK usually pass a tube and aspirate whatever food or fluid is in the baby's stomach before they give a feed. The idea is to ensure that the stomach is not overdistended with food and prevent the baby vomiting or, worse, aspirating food into the lungs. However, there is little justification for this practice. It is rarely done in many other countries. It may not be pleasant for the child and perhaps is unnecessary.

Some experts have suggested that the policy should be evaluated in a randomised controlled trial. This would mean allocating a large number of children at random to either have the stomach aspirated before feeds, or not. Such a trial would be a major undertaking and we are unsure if parents or staff would be willing to allow children to participate.

The aim of this study was to see if it is possible to conduct such a large trial in the UK. Two surveys (of 119 units) showed us that regularly measuring the stomach contents when starting and increasing feeds is common practice for both newborn and older children in UK intensive care units. However, in some countries, such as France, this practice is rarely done.

We asked 31 parents and 51 health-care professionals about a future study. Overall, parents were supportive of a trial if it was explained to them well by a knowledgeable and caring professional, and if they were approached at the right time. Some concerns were expressed about not picking up complications early if gastric residual volume was not measured.

Health-care professionals were also mainly positive about a future trial, but mentioned similar concerns about not picking up complications early and the difficulty of changing a long-standing routine practice. Parents suggested study outcomes that were important to them. These, along with other outcomes, were voted on in a further survey of 106 professionals and at face-to-face meetings involving 41 participants. Overall, our findings suggest that a trial is feasible to perform and acceptable to parents.

However, because of differences in both treatments and important outcomes between children's intensive care units and newborn baby intensive care units, two trials would be needed, one in each type of intensive care unit.

These two trials will test whether or not the benefits of not measuring gastric residual volume (e.g. improved calorie intake) outweigh the potential harms (e.g. delayed diagnosis of complications).

© Queen's Printer and Controller of HMSO 2020. This work was produced by Tume *et al.* under the terms of a commissioning contract issued by the Secretary of State for Health and Social Care. This issue may be freely reproduced for the purposes of private research and study and extracts (or indeed, the full report) may be included in professional journals provided that suitable acknowledgement is made and the reproduction is not associated with any form of advertising. Applications for commercial reproduction should be addressed to: NIHR Journals Library, National Institute for Health Research, Evaluation, Trials and Studies Coordinating Centre, Alpha House, University of Southampton Science Park, Southampton SO16 7NS, UK.

### **Health Technology Assessment**

ISSN 1366-5278 (Print)

ISSN 2046-4924 (Online)

Impact factor: 3.819

Health Technology Assessment is indexed in MEDLINE, CINAHL, EMBASE, the Cochrane Library and Clarivate Analytics, Science Citation Index.

This journal is a member of and subscribes to the principles of the Committee on Publication Ethics (COPE) (www.publicationethics.org/).

Editorial contact: journals.library@nihr.ac.uk

The full HTA archive is freely available to view online at www.journalslibrary.nihr.ac.uk/hta. Print-on-demand copies can be purchased from the report pages of the NIHR Journals Library website: www.journalslibrary.nihr.ac.uk

#### Criteria for inclusion in the Health Technology Assessment journal

Reports are published in *Health Technology Assessment* (HTA) if (1) they have resulted from work for the HTA programme, and (2) they are of a sufficiently high scientific quality as assessed by the reviewers and editors.

Reviews in *Health Technology Assessment* are termed 'systematic' when the account of the search appraisal and synthesis methods (to minimise biases and random errors) would, in theory, permit the replication of the review by others.

#### **HTA programme**

Health Technology Assessment (HTA) research is undertaken where some evidence already exists to show that a technology can be effective and this needs to be compared to the current standard intervention to see which works best. Research can evaluate any intervention used in the treatment, prevention or diagnosis of disease, provided the study outcomes lead to findings that have the potential to be of direct benefit to NHS patients. Technologies in this context mean any method used to promote health; prevent and treat disease; and improve rehabilitation or long-term care. They are not confined to new drugs and include any intervention used in the treatment, prevention or diagnosis of disease.

The journal is indexed in NHS Evidence via its abstracts included in MEDLINE and its Technology Assessment Reports inform National Institute for Health and Care Excellence (NICE) guidance. HTA research is also an important source of evidence for National Screening Committee (NSC) policy decisions.

#### This report

The research reported in this issue of the journal was funded by the HTA programme as project number 16/94/02. The contractual start date was in April 2018. The draft report began editorial review in October 2019 and was accepted for publication in January 2020. The authors have been wholly responsible for all data collection, analysis and interpretation, and for writing up their work. The HTA editors and publisher have tried to ensure the accuracy of the authors' report and would like to thank the reviewers for their constructive comments on the draft document. However, they do not accept liability for damages or losses arising from material published in this report.

This report presents independent research funded by the National Institute for Health Research (NIHR). The views and opinions expressed by authors in this publication are those of the authors and do not necessarily reflect those of the NHS, the NIHR, NETSCC, the HTA programme or the Department of Health and Social Care. If there are verbatim quotations included in this publication the views and opinions expressed by the interviewees are those of the interviewees and do not necessarily reflect those of the authors, those of the NHS, the NIHR, NETSCC, the HTA programme or the Department of Health and Social Care.

© Queen's Printer and Controller of HMSO 2020. This work was produced by Tume *et al.* under the terms of a commissioning contract issued by the Secretary of State for Health and Social Care. This issue may be freely reproduced for the purposes of private research and study and extracts (or indeed, the full report) may be included in professional journals provided that suitable acknowledgement is made and the reproduction is not associated with any form of advertising. Applications for commercial reproduction should be addressed to: NIHR Journals Library, National Institute for Health Research, Evaluation, Trials and Studies Coordinating Centre, Alpha House, University of Southampton Science Park, Southampton SO16 7NS, UK.

Published by the NIHR Journals Library (www.journalslibrary.nihr.ac.uk), produced by Prepress Projects Ltd, Perth, Scotland (www.prepress-projects.co.uk).

### Editor-in-Chief of Health Technology Assessment and NIHR Journals Library

Professor Ken Stein Professor of Public Health, University of Exeter Medical School, UK

### NIHR Journals Library Editors

**Professor John Powell** Chair of HTA and EME Editorial Board and Editor-in-Chief of HTA and EME journals. Consultant Clinical Adviser, National Institute for Health and Care Excellence (NICE), UK, and Senior Clinical Researcher, Nuffield Department of Primary Care Health Sciences, University of Oxford, UK

**Professor Andrée Le May** Chair of NIHR Journals Library Editorial Group (HS&DR, PGfAR, PHR journals) and Editor-in-Chief of HS&DR, PGfAR, PHR journals

**Professor Matthias Beck** Professor of Management, Cork University Business School, Department of Management and Marketing, University College Cork, Ireland

Dr Tessa Crilly Director, Crystal Blue Consulting Ltd, UK

Dr Eugenia Cronin Senior Scientific Advisor, Wessex Institute, UK

Dr Peter Davidson Consultant Advisor, Wessex Institute, University of Southampton, UK

Ms Tara Lamont Director, NIHR Dissemination Centre, UK

Dr Catriona McDaid Senior Research Fellow, York Trials Unit, Department of Health Sciences, University of York, UK

Professor William McGuire Professor of Child Health, Hull York Medical School, University of York, UK

Professor Geoffrey Meads Professor of Wellbeing Research, University of Winchester, UK

Professor John Norrie Chair in Medical Statistics, University of Edinburgh, UK

**Professor James Raftery** Professor of Health Technology Assessment, Wessex Institute, Faculty of Medicine, University of Southampton, UK

Dr Rob Riemsma Reviews Manager, Kleijnen Systematic Reviews Ltd, UK

Professor Helen Roberts Professor of Child Health Research, UCL Great Ormond Street Institute of Child Health, UK

Professor Jonathan Ross Professor of Sexual Health and HIV, University Hospital Birmingham, UK

**Professor Helen Snooks** Professor of Health Services Research, Institute of Life Science, College of Medicine, Swansea University, UK

Professor Ken Stein Professor of Public Health, University of Exeter Medical School, UK

**Professor Jim Thornton** Professor of Obstetrics and Gynaecology, Faculty of Medicine and Health Sciences, University of Nottingham, UK

Professor Martin Underwood Warwick Clinical Trials Unit, Warwick Medical School, University of Warwick, UK

Please visit the website for a list of editors: www.journalslibrary.nihr.ac.uk/about/editors

Editorial contact: journals.library@nihr.ac.uk